Author’s response to reviews

Title: Prognostic factors for survival in patients with malignant pleural effusions. What is considered important for treatment?

Authors:

Mauro M Zamboni (mauro.zamboni@gmail.com)
Cyro T Silva Jr (ctsilvajunior@predialnet.com.br)
Rodrigo Baretta (rodrigobaretta@hotmail.com)
Edson T Cunha (edsontoscano@gmail.com)
Gilberto P Cardoso (gilpcard@globo.com)

Version: 5 Date: 26 February 2015

Author’s response to reviews: see over
Ms Hazel Joyce Delos Santos  
BMC Pulmonary Medicine  

February, 26th, 2015  

Dear Ms Delos Santos:  

I am pleased to submit an original Research Article entitled “Prognostic factors for survival in patients with malignant pleural effusion.” by Mauro Musa Zamboni; Cyro Teixeira da Silva Jr; Rodrigo Baretta; Edson Toscano Cunha; and Gilberto Perez Cardoso for consideration for publication in the BMC Pulmonary Medicine.  

As requested by the Editor I included all the authors in the Authors’ contribution section.  

The authors would like to acknowledge the contributions, comments and suggestions from the Editor. All of them were of great help in drawing up our paper.  

Thank you for your consideration!  

Sincerely,  

Mauro M. Zamboni